Author's response to reviews

Title: Acute abdomen and renal failure due to bilateral renal infarction in a woman: a case report

Authors:

khaireddine bouassida (khayri14@hotmail.com)
wissem hmida (khayri14@hotmail.com)
amira zairi (khayri14@hotmail.com)
adnen hidoussi (khayri14@hotmail.com)
faouzi mallat (khayri14@hotmail.com)
mehdi jaidane (khayri14@hotmail.com)
khaled hani (khayri14@hotmail.com)
faouzi ali mosbah (khayri14@hotmail.com)

Version: 6 Date: 2 March 2012

Author's response to reviews:

We would think associate editor for their informative comments with regards to our article. Please find below responses.

We try to make the presentation more coherent so the whole paper was edited by an expert in English. The paper was checked over by native English speaker